Author’s response to reviews

Title: Toward Criteria for Pragmatic Measurement in Implementation Research and Practice: A Stakeholder-Driven Approach Using Concept Mapping

Authors:

Byron Powell (bjpowell@unc.edu)
Cameo Stanick (cameo.stanick@gmail.com)
Heather Halko (heather.halko@umconnect.umt.edu)
Caitlin Dorsey (dorsey.c@ghc.org)
Bryan Weiner (bjweiner@uw.edu)
Melanie Barwick (melanie.barwick@sickkids.ca)
Laura Damschroder (Laura.Damschroder@va.gov)
Michel Wensing (Michel.Wensing@med.uni-heidelberg.de)
Cara Lewis (lewis.cc@ghc.org)

Version: 2 Date: 07 Sep 2017

Author’s response to reviews:

Dear Dr. Straus,

We have removed references to the conference paper, and have cited the paper that is under review instead. Thank you!

Best,

Byron